Author’s response to reviews

Title: Ursodeoxycholic acid augmentation in treatment-refractory schizophrenia: A case report

Authors:
Mohsen Khosravi (dr_khosravi2016@yahoo.com)

Version: 2 Date: 19 Jun 2020

Author’s response to reviews:

Dear Reviewers,
I would like to thank you for your valuable comments. The revisions you asked for are listed below:

Reviewer #2
1. Based on your recommendation, the discussion and conclusions sections were rewritten. Due to the novelty of the findings, it was not possible to further explain the feasible therapeutic mechanisms of UDCA.
2. TSR was changed to TRS.